Author's response to reviews

Title: Prolactinomas, Cushing's disease and acromegaly: debating the role of medical therapy for secretory pituitary adenomas

Authors:

Beverly MK Biller (bbiller@partners.org)
Annamaria Colao (colao@unina.it)
Stephan Petersenn (stephan.petersenn@uni-due.de)
Vivien S Bonert (Vivien.Bonert@cshs.org)
Marco Boscaro (m.boscaro@ao-umbertoprimo.marche.it)

Version: 4 Date: 12 May 2010

Author's response to reviews:

As requested, the summary text box has been removed and the information that was in the text box is now within the manuscript text, the references are in sequence, the article has been reformatted for a review-type article, a list of authors' email addresses has been made (without underlining), the word count etc has been deleted from the title page, the references have been formatted to contain up to 30 author names and highlighting has been removed. Permission has been gained from the copyright holders for the three figures.